

## ANSWERING REVIEWERS

June 15, 2015



**Title:** Esophageal granular cell tumors: Case report and literature review

**Author:** Wang Hong-qun, Liu Ai-jun

**Name of Journal:** *World Journal of Gastrointestinal Oncology*

**ESPS Manuscript NO:** 17749

Dear Editor,

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

(1) supplementing some new literature

(2) having the paper read and corrected by a native speaking specialist

(3) having described the type of endoscopic mucosal resection performed

(4) deleting some the endoscopic description of the gastric mucosa appearance

(5) In case 3: Why not USE was performed prior to EMR? Because our hospital hasn't endoscope ultrasonography, a physician estimated the mass motion and position though touching the mass with biopsy forceps. If the mass the mucosa is  $\leq 3$  cm in diameter, good activity, and located in the mucosa and submucosa, the doctor may use endoscopic mucosal resection.

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *Gastrointestinal Oncology*.

Sincerely yours,

*Wang Hong-qun*

Wang Hong-qun, Attending doctor, Master,

*Department of Pathology*

*The Third People's Hospital, Hefei*

*Anhui province, China*

Fax: +86-551-63666145

E-mail: [whqyy@sina.com](mailto:whqyy@sina.com)